Author's response to reviews

Title: Seizures as the first manifestation of chromosome 22q11.2 deletion syndrome in a 40-year-old man.

Authors:

Adriano R Tonelli (Adriano.Tonelli@ht.msu.edu)
Kalyan Kosuri (Kalyan.Kosuri@ht.msu.edu)
Sainan Wei (weisaina@msu.edu)
Davoren Chick (Davoren.Chick@hc.msu.edu)

Version: 2 Date: 17 April 2007

Author’s response to reviews:

Please separate the Conclusions and Discussion sections - this is not journal style.

Please detail whether written patient consent was obtained within an Acknowledgements section. All sources of funding should also be declared in the Acknowledgements section.

Please insert the words "Case report:" into the title of the manuscript.

Please remove the cover letter from the end of the manuscript.

Please provide a Figure title and legend section after the references list - the figures themselves should not include the title or legend.

DONE!